

REVIEW HISTORY

Manuscript Title: *Observations on Self-resilience in Italian-American Identity: Education and Sacrifice in Helen Barolini's Umbertina*

Submitted: 22 January, 2025

Accepted: 19 May, 2025

Published: 24 May, 2025

Corresponding Author: Matteo Cacco

Email: mcacco1@uni-koeln.de

Doi: <http://dx.doi.org/10.19044/llc.v12no1a3>

PEER REVIEW

Reviewer 1: Olga Dabrowska-Cendrowska

Reviewer 2: Wafaa Abdel-Kader Mostafa Hussein